Author’s response to reviews

Title: Predicting life expectancy with a long short-term memory recurrent neural network using electronic medical records

Authors:

Merijn Beeksma (m.t.beeksma@let.ru.nl)

Suzan Verberne (s.verberne@liacs.leidenuniv.nl)

Antal van den Bosch (antal.van.den.bosch@meertens.knaw.nl)

Enny Das (h.das@let.ru.nl)

Iris Hendrickx (iris@i-hx.nl)

Stef Groenewoud (stef.groenewoud@radboudumc.nl)

Version: 1 Date: 26 Jun 2018

Author’s response to reviews:

Dear editorial board,

Please find a revised version of our submission. We addressed the following comments:

"1. Ethics Approval:

Please confirm whether your study was submitted to and approved by your institutional ethics committee and include a statement to this effect in your Declarations section. Please also ensure that the full name of your ethics committee is included in this statement. If the need for ethics approval was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

2. Submission Guidelines:

Please ensure that your manuscript is structured according to our submission guidelines. The 'Results' and 'Discussion' should be provided as separate sections. For more information, please
see: https://bmcmedinformdecismak.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

3. Figure Legends:

Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

4. Authors' Contributions:

Please list each individual author and their contributions in the Authors’ Contributions section (e.g. Authors AB, CD, EF and GH participated in data acquisition, analysis, and interpretation). Currently the individual contributions of several authors are missing.

5. Funding:

If no funding was obtained for your study we still require this section to be included with the statement “No funding was obtained for this study”.

We thank you for the helpful suggestions, and hope that the paper now has the correct format to be sent to reviewers.

On behalf of all authors,

Best regards,

Merijn Beeksma